Author’s response to reviews

Title: The Trends in Prostate Specific Antigen Usage Amongst United Kingdom Urologists, a Questionnaire Based Study.

Authors:

Helena P Burden Miss (helena.burden@bristol.ac.uk)
Chris R Davis Dr (ChrisDavis959@hotmail.com)
Sophie Tate Miss (St3284@bristol.ac.uk)
Raj Persad Mr (rajpersad@bristolurology.com)
Chris H Holmes Dr (chris.holmes@bristol.ac.uk)
Kate Whittington Dr (k.whittington@bristol.ac.uk)

Version: 2 Date: 19 February 2008

Author’s response to reviews: see over
Manuscript number: 1595296551176069
Re: The Trends in Prostate Specific Antigen Usage Amongst United Kingdom Urologists, a Questionnaire Based Study.
Authors: Helena P Burden Miss, Chris R Davis Dr, Sophie Tate Miss, Raj Persad Mr, Chris H Holmes Dr and Kate Whittington Dr.

19.02.2008

Dear Ann,

Sorry for the delay.

I have changed the methods section as suggested to include data regarding the two rounds. My changes are highlighted in yellow. I can confirm that the exclusion criteria remained the same for both rounds, and that the responders in the first round would all fit into the inclusion criteria.

I have attached the revised manuscript and look forward to hearing from you.

Best wishes,
Helena Burden